Author’s response to reviews

Title: Development and Psychometric Assessment of Triage Nurses' Professional Capability Questionnaire in Emergency Department

Authors:
Mostafa Bijani (bizhani_mostafa@yahoo.com)
Mahnaz Rakhshan (rakhshanm@sums.ac.ir)
Mohammad Fararouei (fararoeei@yahoo.com)
Camellia Torabizadeh (torabik@sums.ac.ir)

Version: 2 Date: 14 Aug 2020

Author’s response to reviews:

Dear Editor

Firstly, we would like to thank the editor for the careful reading of the script and helpful comments. Comments were made corrections as follows:

♦ The abstract and methods were revised to minimize overlap.
♦ The order of authors in the system is the same as in the title page of the manuscript.
♦ List of abbreviations provided in the Declarations.
♦ The manuscript upload as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. That all figures, tables and additional-supplementary files are cited within the text.
♦ All figures/tables and supplementary files are cited within the text.
♦ Response to the reviewers/Cover letter, removed from supplementary files

Best wishes

Dr, Torabizadeh